Author’s response to reviews

Title: Dislocation of fibular head. An unusual sports injury: A case report

Authors:

riaz ahmad (riazkanth@yahoo.co.uk)

Version: 4 Date: 6 March 2008

Author’s response to reviews:

6 March 2008

MS: 1942162242168811
Dislocation of fibular head. An unusual sports injury: A case report

Dear Sir/ Madam

Kindly find attached the revised manuscript along with the changes suggested by the reviewer.

The reviewer’s comments have been addressed as following;

1. The side of injury has been mentioned in the text as ‘right’ in the second sentence of the case presentation.
2. The radiogram of the knee has been included with the manuscript as figure 2.
3. The operative findings have been included in the last paragraph of the case presentation as ‘There was no bony or ligamentous injury’.
4. The disjointed beginning of the discussion has been corrected.

Regards

Riaz Ahmad